Author's response to reviews

Title: How many preterm births in England are due to excision of the cervical transformation zone? Nested case control study

Authors:

Rekha Wuntakal (wrekhaw@gmail.com)
Alejandra Castanon (a.castanon@qmul.ac.uk)
Rebecca Landy (r.landy@qmul.ac.uk)
Peter D Sasieni (p.sasieni@qmul.ac.uk)

Version: 5
Date: 27 August 2015

Author's response to reviews: see over
We have now addressed the comment below, and hope this is now satisfactory.
Best wishes,
Rebecca Landy

This is a follow up to the email sent yesterday. In addition to the editorial points mentioned in our previous email, we kindly ask that you clarify a point regarding authorship:

In the title page, the list of authors includes the statement “on behalf of the PaCT Investigators”. It is unclear whether this applies to author P. Sasieni or to all authors listed. If it refers to a specific author, please explain why this author was chosen to represent the group.

Please bear in mind that a collaboration group which contributed to authorship can and should be listed as an author in the title page and its contribution listed in the Authors’ Contribution section. Individuals within the group can then be specified in the Acknowledgments section, as is already the case in your manuscript. Please refer to the Instructions for Authors for details regarding authorship: http://www.biomedcentral.com/bmcpregnancychildbirth/authors/instructions/researcharticle

Should the group not meet the criteria for authorship, it should be removed from the title page and listed only in the Acknowledgments.

The PaCT investigators should be acknowledged as non-author contributors. They have all been involved in the collection of data for the study and/or in the steering group for the study. All named authors have written this manuscript on behalf of the PaCT investigators. As this is the case we have removed them from the title page and list them only in the acknowledgements.